Author's response to reviews

Title: Prosthetic valve endocarditis caused by Pseudomonas luteola

Authors:

Jean-Paul Casalta (jean-paul.casalta@wanadoo.fr)
Pierre-Edouard Fournier (Pierre-Edouard.Fournier@medecine.univ-mrs.fr)
Gilbert Habib (gilbert.habib@ap-hm.fr)
Alberto Riberi (alberto.riberi@ap-hm.fr)
Didier Raoult (didier.raoult@medecine.univ-mrs.fr)

Version: 4 Date: 20 September 2005

Author’s response to reviews:

Dear Editor,

Attached, please find the revised version of our manuscript entitled "Prosthetic valve endocarditis caused by Pseudomonas luteola" (MS# 1882831096074374). The manuscript was revised following the editor’s comments as follows:

1) Line numbers were removed

None of the authors has any conflict of interest related to this research.
All authors have participated in the study and concur with the submission and subsequent revisions submitted by the corresponding author.

Yours, sincerely,

Jean-Paul CASALTA